Author’s response to reviews

Title: Prevalence of Primary Dysmenorrhea, Its Intensity, Impact and Associated Factors among Female Students’ at Gondar Town Preparatory School, Northwest Ethiopia

Authors:
Abere Azagew (wabere@ymail.com)
Destaye Kassie (dstgd32@gmail.com)
Tarkie Wale (robaname@gmail.com)

Version: 1 Date: 31 May 2018

Author’s response to reviews:

1. Please could you clarify the age of the youngest participant included in your study?

Response:
The age of the study participants range from 17-23, so the youngest age in this study is 17 years.

2. In your “ethical approval and consent to participate” section of your declarations please state if consent to participate was obtained from the parents/guardians of the minors included in this study (minors are considered anyone under the age of 16). If no participants were ages under 16, then please disregard this request.

Response: thank you but No study participants’ under the age of 16 years.

3. In your “ethical approval and consent to participate” section of your declarations please confirm if an ethics committee approved the procedure for verbal consent, and why they felt the need for written consent was not necessary for this study. Please also detail how you documented the consent.
Response: The ethics committee approved the proposal and letter of ethical clearance was given for the authors to proceed the study. The author used verbal informed consent rather than written consent for the study participants to be included in the study. The objective as well as the benefit of the study was explained to the study participants before administering the questionnaire and any study participant can quit from the study if they faced any problem regarding the study.

4. Please provide figure legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Response: Thank you: Corrected